Math 153 Spring 2005 R. Schultz

UPDATED GENERAL INFORMATION — APRIL 27, 2005

Here is the fifth homework assignment, which is due in class on Wednesday, May 4, 2005. As usual, “Burton” refers to the course text by Burton.

Burton, p. 220: 13

Burton, p. 226: 1b, 2bd

It would also be worthwhile to try a few of the other problems on pages 220-221 of Burton and look at the book’s solutions afterwards. The same applies to the portions of Exercises 1–2 on page 226 that were not assigned.

STATISTICS FROM THE FIRST MIDTERM EXAMINATION

The cutoff scores are as follows: A — 75 B — 51 C — 39 D — 27

The median score was 55.

Appeals regarding the grading of this examination must be submitted by the end of class on Friday, May 6. Written comments should be placed on the examination indication the problems to be reconsidered. BRIEF and OBJECTIVE statements about specific issues may be included.